
Quickie status - 10 Mar
The field crew waited out the rain this morning and then rushed to install s1 in the afternoon.  Everything appears to running, except the barometer, no 
doubt the 7E1 problem.

Thus, we now have reporting s1, s4, s8, s10, s14, s15.

Also, s17 briefly came in this afternoon (22:29 - 22:37 UTC). Vmote values were generally reasonably high, indicating that the station has power.

Ott: all okay

TRH: s10 died today, s8 reporting fan not working, others okay

P: s1 not reporting – probably 7E1 issue (but can't log in to fix), others okay

CSAT: all okay

EC150: s8 bad, others okay

Gill 2D: all okay (but need to add to qctables)

Rad: all okay 

Soils: all okay

Victron: s17 not reporting – probably usb rules setting (but can't log in to fix), others okay

mote: all okay, changed "sn" setting on s4 to report serial numbers
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